Author's response to reviews

Title: Does Korea have hidden female smokers? Smoking rates according to self-reported surveys versus biological markers

Authors:

Myung Bae Park (parklove5004@naver.com)
Eun Woo Nam (ewnam@yonsei.ac.kr)
Kyeong Soo Hong (suva@khealth.or.kr)
Chun-Bae KIM (kimcb@yonsei.ac.kr)

Version: 4 Date: 15 April 2014

Author's response to reviews:

This revised manuscript contains ethics statement (IRB approval) descriptions of the study method. And, title page includes all author information, and the full postal address of the correspondence author.